Author's response to reviews

Title: Development of a complex intervention to test the effectiveness of peer support in type 2 diabetes

Authors:

Gillian Paul (gpaul@tcd.ie)
Susan M Smith (susmith@tcd.ie)
David Whitford (dwhitford@rcsi.ie)
Fergus O'Kelly (feokelly@tcd.ie)
Tom O'Dowd (todowd@tcd.ie)

Version: 6 Date: 7 August 2007

Author's response to reviews:

Department of Public Health and Primary Care,
Trinity College Dublin,
07/08/07

Dear Mr. Hopkins,

Thank you for your email of 31/08/07 regarding our submission the BMC Health Services Research (1975486050135708 Resubmission 4 Development of a complex intervention to test the effectiveness of peer support in type 2 diabetes).

I attach a corrected version of the manuscript. Unfortunately we are still waiting for a statement of permission to reproduce Figure 2. This delay is due to the holiday period and the relevant people being on leave. I will contact you as soon as I receive this statement.

Please do not hesitate to contact me if you have any further queries.

Yours sincerely,

Gillian Paul